Author's response to reviews

Title: Primary muscular cystic hydatidosis: case report

Authors:

Sonia Vicidomini (svicidomini@hotmail.com)
Gabriella Cancrini (gabriella.cancrini@uniroma1.it)
Simona Gabrielli (simona.gabrielli@uniroma1.it)
Riccardo Naspetti (naspettir@ao-careggi.toscana.it)
Alessandro Bartoloni (bartoloni@unifi.it)

Version: 4 Date: 8 March 2007

Author’s response to reviews: see over
Point-to-point response to the referee comments:

Reviewer: Enrico Brunetti
We accepted all the suggested changes. As for Conclusions-line 21, we modified the text of the case report section describing what was the histological report.

Reviewers: Hugo Garcia and Fabrizio Bruschi
No changes were requested

Reviewer: Nefise Cagla Tarhan
We put pre contrast and after contrast of the T1-weighted images and modified the legend accordingly.